Author's response to reviews

Title: Intensified thermal management for patients undergoing transcatheter aortic valve implantation (TAVI)

Authors:

Ivo F Brandes (ibrande@gwdg.de)
Marc Jipp (marcjipp@gmx.de)
Aron F Popov (popov@med.uni-goettingen.de)
Ralf Seipelt (rseipelt@med.uni-goettingen.de)
Michael Quintel (mquintel@med.uni-goettingen.de)
Anselm Bräuer (abraeue@gwdg.de)

Version: 3 Date: 18 September 2011

Author's response to reviews: see over
Formatting changes to be made:
---------------------------------  
Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs).
The author list has been formatted accordingly.

Affiliations: It is important that every institution should be included as a separate affiliation with its own number. Different departments of the same building should be given as separate affiliations. Authors should then be linked to their affiliations using the relevant number in superscript after their name. Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.
The affiliations have been changed as requested.

'Conflicts of Interests': Please change the title of this section to 'Competing Interests' and move it so it comes directly after the conclusions section.
The section has been renamed and moved to the proper position within the manuscript.

Abstract: The abstract in the manuscript and the abstract in the online submission system must be identical. This includes section headings. Please ensure both of the abstracts are identical.
The abstracts are now identical.

Tables: we are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold.
The table has been edited accordingly.

Authors' contributions: Please place the Authors contributions section towards the end of the manuscript text, directly preceding the Acknowledgements.
Please use the following format as an example, including using initials to refer to each AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. Please indicate that all authors read and approved the final manuscript.
The authors’ contributions have been changed to match the journal’s format.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Tracking: Please remove all forms of tracking and ensure that all text in the manuscript is black.
All tracking has been removed and all text is black.

Short title: Please remove this from the title page.
The short title has been removed from the title page.
Funding: Please move the statement of funding to the Acknowledgements section and remove it from the title page
The statement has been moved to the end of the manuscript.

Tables: Please remove any visible vertical lines present in any tables in the manuscript. Also please place table numbers above their respective table and place table legends below them.
The table has been edited as requested.

References: Please remove 'table of contents.' given at the end of reference 11.
Done.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here:
http://www.biomedcentral.com/info/ifora/figures

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
The typography has been checked.

Thank you
-----------------------------